Author’s response to reviews

Title: Analysis of Cardiovascular Mortality, Bleeding, Vascular and Cerebrovascular Events in Patients with Atrial Fibrillation vs. Sinus Rhythm Undergoing Transfemoral Transcatheter Aortic Valve Implantation (TAVR)

Authors:

Joerg Herold (Joerg_herold@hotmail.com)
Vasiliki Herold-Vlanti (v1985@windowslive.com)
Mohammad Sherif (mohammad.sherif@med.uni-rostock.de)
Blerim Luani (blerim.Luani@med.ovgu.de)
Klaus Bonaventura (kbonaventura@klinikumevb.de)
Christin Breyer (christinbreyer@gmx.de)
Ruediger Braun-Dullaeus (r.Braun-Dullaeus@med.ovgu.de)

Version: 3 Date: 17 Nov 2017

Author’s response to reviews:

Complete Revision letter will be send separately

Revision: Manuscript number: BCAR-D-17-00421R3

Dear Prof. Barbero, Prof. Barbetta and Editorial Team,

We appreciate the comments from the reviewers, which allowed us to substantially improve the manuscript BCAR-D-17-00421R3. We have responded to all the suggestions from the editor and the reviewers, and we have made all the suggested corrections to the text.

We have accepted the modifications by the Science Editor to ensure our manuscript complies with the BMC formatting standard.

Please find included our manuscript by Herold et al. entitled "Analysis of Cardiovascular Mortality, Bleeding, Vascular and Cerebrovascular Events in Patients with Atrial Fibrillation vs.
Sinus Rhythm Undergoing Transfemoral Transcatheter Aortic Valve Implantation (TAVR)” to be considered for publication in the journal “BMC Cardiovascular Disorders”.

Best regards,

Priv. Doz. Dr. med. habil. Joerg Herold